Author's response to reviews

Title: Evaluating housing quality, health and safety using an Internet-based data collection and response system: a cross-sectional study

Authors:

Mari Turunen (mari.turunen@thl.fi)
Ari Paanala (ari.paanala@thl.fi)
Juha Villman (juha.villman@thl.fi)
Aino Nevalainen (aino.nevalainen@thl.fi)
Ulla Haverinen-Shaughnessy (ulla.haverinen-shaughnessy@thl.fi)

Version: 3 Date: 5 November 2010

Author's response to reviews: see over
Dear Editor-in-Chief,

Attached please find a revised version (version 3) of the manuscript titled “Evaluating housing quality, health and safety using an Internet based data collection and response system”.

We are grateful for the comments. All the changes made to the manuscript are marked with track changes.

The language check has been performed twice (by two native-English speakers).

Sincerely,
Mari Turunen, Researcher
Master of Science, PhD. student
E-mail: mari.turunen@thl.fi